Author's response to reviews

Title: Circulating tumor cells and survival of patients with gastric cancer: a meta-analysis

Authors:

  Zhen-yu Zhang (zhangsurgery@qq.com)
  Zhen-ling Dai (circulating@qq.com)
  Xiao-wei Yin (translational@qq.com)
  Shu-heng Li (prilo@qq.com)
  Shu-ping Li (microfluidic@qq.com)
  Hai-yan Ge (gesurgery@qq.com)

Version: 2 Date: 9 May 2014

Author's response to reviews: see over
Dear editors,

Thank you for your letter on 9 May 9, 2014, reminding us to revise our manuscript. The following are our specific response to the comments.

Comments: #1 Line Numbering

Response: Thank you. We have checked the manuscript thoroughly and continuously numbered each line of the word document of the manuscript accordingly. The revised manuscript is also uploaded.

Thank you.

Best regards,

Hai-yan Ge